Author's response to reviews

Title: Neck/shoulder pain in adolescents is not related to the level or nature of self-reported physical activity or type of sedentary activity in an Australian pregnancy cohort

Authors:

Andrew M Briggs (A.Briggs@curtin.edu.au)
Leon M Straker (L.Straker@curtin.edu.au)
Natasha L Bear (N.Bear@curtin.edu.au)
Anne J Smith (Anne.Smith@curtin.edu.au)

Version: 3 Date: 13 July 2009

Author's response to reviews: see over
BioMed Central Editorial Production Team

BMC Musculoskeletal Disorders

Re: submission MS 2123171509263076

“Neck/shoulder pain in adolescents is not related to the level or nature of physical activity or the type of sedentary activity in an Australian pregnancy cohort”

Dear Sir/Madam,

Thank you for your email indicating that minor formatting changes are required to the abovementioned manuscript prior to publication. On behalf of the co-authors, I am pleased to make the changes as listed below.

1. The city for each affiliation has been included
2. Author qualifications have been removed
3. The header has been removed
4. Track changes have been removed
5. Keywords have been removed
6. Underlining has been removed and replaced with a boldface font where appropriate
7. The additional file containing a duplicate manuscript has been removed
8. The manuscript has been checked for typographical errors

Sincerely

Andrew Briggs